Author's response to reviews

Title: What they fill in today, may not be useful tomorrow: Lessons learned from studying Medical Records at the Women hospital in Tabriz, Iran

Authors:

Faramarz Pourasghar (Faramarz.pourasghar@ki.se)
Hossein Malekafzali (afzali@hbi.ir)
Alireza Kazemi (dr_alireza_kazemi@yahoo.com)
Johan Ellenius (Johan.Ellenius@ki.se)
Uno GH Fors (Uno.Fors@ki.se)

Version: 5 Date: 25 April 2008

Author's response to reviews:

Dear editor, we have now also made the last formatting changes suggested

Uno Gh Fors